Author’s response to reviews

Title: Does alcohol use have a causal effect on HIV incidence and disease progression? A review of the literature and a modeling strategy for quantifying the effect

Authors:

Jürgen Rehm (jtrehm@gmail.com)
Charlotte Probst (charlotte.probst@camh.ca)
Kevin Shield (kevin.david.shield@gmail.com)
Paul Shuper (paul.shuper@camh.ca)

Version: 2 Date: 02 Dec 2016

Author’s response to reviews:

RTR